Author's response to reviews

Title: Diffuse leptomeningeal gliomatosis initially presenting with intraventricular hemorrhage: a case report and literature review

Authors:

Min Zhu (zhumin1@126.com)
Junjun Zheng (zjj259@163.com)
Yuanzhao Zhu (846366855@qq.com)
Hui Wan (ncwanhui@sina.com)
Yuchen Wu (wuyuchen52@163.com)
Daojun Hong (hongdaojun@hotmail.com)

Version: 5
Date: 29 April 2015

Author's response to reviews:

Dear editors:

I am glad to submit the revised manuscript to you. We made a great effort to improve the English language in the revised manuscript. We made a point-by-point response to the concerns raised by reviewer.

Reply to Professor Joseph Burns (This revision represents a substantial improvement over the first draft. Overall it is an interesting case report that should be of interest to this journal's readers. While improved compared to the previous draft, it still needs English language editing.)

Reply: We appreciated your academic opinions. We carefully revised the English language. Meanwhile, an extensive English language editing was made by a native speaker.

Yours sincerely,

Dr. Daojun Hong
Department of Neurology, The first affiliated hospital of Nanchang University
Yong Wai Zheng Street 17#, Nanchang, 330006, P.R.China
Email: hongdaojun@hotmail.com, Telephone: 86-791-8869-2511